Author’s response to reviews

Title: A functional polymorphism in the promoter of miR-17-92 cluster is associated with decreased risk of ischemic stroke

Authors:
Huato Huang (huatuohuang@126.com)
Guijiang Wei (252787634@qq.com)
Chunfang Wang (578457776@qq.com)
Yulan Lu (yulanlu15@163.com)
Chunhong Liu (249065242@qq.com)
Rong Wang (714836941@qq.com)
Xiang Shi (563998332@qq.com)
Jun Yang (junyang1232@163.com)
Yesheng Wei (yeshengwei22@163.com)

Version: 3 Date: 19 Sep 2019

Author’s response to reviews:

Dear editor and reviewer,

Thank you very much for your letter on 15 Sep 2019 enclosing comments on our manuscript entitled “A functional polymorphism in the promoter of miR-17-92 cluster is associated with decreased risk of ischemic stroke” (MGNM-D-18-00369R2). We have revised the manuscript according to the comments and point by point responses to the comments are listed below. We hope that the revised version of the manuscript is now acceptable for publication in BMC Medical Genomics. We are looking forward to hearing from you soon.

Best wishes,

Yours sincerely,
Huatro Huang; Guijiang Wei; Chunfang Wang; Yulan Lu; Chunhong Liu; Rong Wang; Xiang Shi; Jun Yang; Yesheng Wei

1. Please ensure the email address for the corresponding author in the manuscript title page matches that of the corresponding author in the submission system.

Response to the Comment: Dear editor and reviewer, we have checked the email address carefully and we can confirm that the email address of the corresponding authors are correct.

2. Please include a 'Declarations' heading above your Declarations.

Response to the Comment: Dear editor and reviewer, 'Declarations' heading has been included as request.

3. Please include a statement in your Funding section describing the role of the funding body in the design of the study, the collection, analysis, and interpretation of data and in writing the manuscript.

Response to the Comment: Dear editor and reviewer, the statement has been added to the funding section. Thanks for your suggestion.

4. Please change the title of your ‘Introduction’ to ‘Background’.

Response to the Comment: Dear editor and reviewer, ‘Introduction’ has been changed to ‘Background’ as request. Thanks for your suggestion.

5. Please change the title of your ‘Materials and Methods’ section to ‘Methods’.

Response to the Comment: Dear editor and reviewer, ‘Materials and Methods’ has been changed to ‘Methods’ as request. Thanks for your suggestion.

6. Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file. Please remove any figures from the main text of the manuscript.

Response to the Comment: Dear editor and reviewer, thanks for your suggestion. Figures will be submitted as a single file as request in the revised version of manuscript and figures in the main text of the manuscript will also be removed.
7. Please provide figure titles and legends under a separate heading of 'Figure Legends' after the References. If figure titles/legends are within the main text of the manuscript, please move them.

Response to the Comment: Dear editor and reviewer, titles and legends have been provided as a separate heading of 'Figure Legends' after the References, and titles/legends within the main text of the manuscript will also be removed from the main text in the revised version of manuscript.

8. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them. This includes simple titles like 'Figure 1'.

Response to the Comment: Dear editor and reviewer, we have checked the image/graphic once again carefully, and there is no titles/legends, or simple titles in any image/graphic we provided.

9. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response to the Comment: Dear editor and reviewer, manuscript will be uploaded as a single, and there will not any tracked changes, comments, highlights, strikethroughs or text in different colours in the main text in the new version of manuscript. Thank you for your suggestion.